185

Open Access Full Text Article

Non-Alcoholic Steatohepatitis Patient Characterization and Real-World Management Approaches in Italy

Elisabetta Bugianesi¹, Luca Miele^{2,3}, Giovanna Donnarumma⁴, Katrine Grau⁵, Mariarosaria Mancuso⁴, Preethy Prasad⁵, Andrea Leith⁶, Victoria Higgins ¹/₁₀

¹Division of Gastroenterology, Department of Medical Sciences, University of Torino, AOU Città della Salute e della Scienza, Torino, Italy; ²Dipartimento di Scienze Mediche e Chirurgiche Addominali ed Endocrino Metaboliche, Fondazione Policlinico Universitario A. Gemelli IRCCS, Rome, Italy; ³Dipartimento di Medicina e Chirurgia Traslazionale, Università Cattolica del Sacro Cuore, Rome, Italy; ⁴Novo Nordisk SpA, Rome, Italy; ⁵Novo Nordisk A/S, Søborg, Denmark; ⁶Adelphi Real World, Bollington, Cheshire, UK

Correspondence: Elisabetta Bugianesi, Email elisabetta.bugianesi@unito.it

Background: Although the estimated prevalence of non-alcoholic steatohepatitis (NASH) in Italy is 4–6%, little is known about patient characteristics and care pathways.

Aim: To describe patient characteristics and management approaches for patients with NASH or suspected NASH in Italy.

Methods: Data were drawn from the Adelphi Real World NASH Disease Specific Programme[™], a cross-sectional survey of endocrinologists and gastroenterologists in Italy from January to March 2018. Physicians completed questionnaires for their next five consecutively consulting patients with NASH or suspected NASH. Analyses were descriptive.

Results: Seventy-six physicians provided data on 380 patients. The mean age was 58.5 ± 11.1 years and the mean body mass index was 31.8 ± 5.5 kg/m². A total of 231 patients (61%) had no/non-advanced fibrosis as evaluated by liver biopsy or non-invasive tests. Common diagnostic assessments were cholesterol, hemoglobin A1c, absence of viral hepatitis, and alcohol assessment. At diagnosis, 87% (n=322/372) and 45% (n=169/372) of patients received an ultrasound and liver biopsy, respectively. Overall, 88% of patients were referred from primary to secondary care. Obesity (81%) and type 2 diabetes (62%) were the most commonly recorded comorbidities, with 70% of patients having ≥ 3 comorbidities. Vitamin E (13%) and GLP-1 receptor agonists (13%) were the most prescribed guideline-recommended treatments for all patients.

Conclusion: Patients with NASH in Italy had high levels of obesity and comorbidities, while diagnosis and treatment frequently were not according to guidelines. Our data show an unmet need for more targeted diagnosis and treatment in Italian patients with NASH, in order to optimize outcomes.

Plain language summary: Fat buildup in the liver, known as fatty liver disease, affects around 4-6% of people in Italy, and can lead to complications if left untreated. However, little is known about how doctors manage people with this disease. Fatty liver disease can only be diagnosed by a liver biopsy, but we found this is only performed in 45% of people, making accurate diagnosis difficult. We also found that 28% of people receive a recommended medication. Overall, increasing doctor awareness of fatty liver medical guidelines may help improve the diagnosis and treatment for people living with fatty liver disease in Italy.

Keywords: NASH diagnosis, NASH treatment, NASH resource utilization

Introduction

Non-alcoholic fatty liver disease (NAFLD) is the most common cause of chronic liver disease worldwide, with approximately 30% of the general adult population potentially affected.¹ In Italy, the prevalence of NAFLD and its progressive form, non-alcoholic steatohepatitis (NASH), has been examined in several studies. A study using vibration-controlled transient elastography (VCTE), conducted in the Palermo region, revealed a NAFLD prevalence of 48%, with

possible advanced fibrosis in 6.5% of cases.² A population-based nested case-control study of all individuals aged ≥ 18 years registered at Italian primary care services revealed a NAFLD prevalence of 9%, with high regional variability.³ A Markov model, based on the adult prevalence of obesity and type 2 diabetes mellitus (T2DM), predicted that the prevalence of NAFLD in Italy will increase from 25.4% in 2016 to 29.5% in 2023; with the corresponding values for NASH being 4.4% and 6.3%, respectively.⁴

Most patients with NAFLD are asymptomatic.⁵ The symptoms of NASH may be non-specific, thus making diagnosis challenging. Symptoms include fatigue, abdominal pain, and sleep difficulties.⁶ A definitive diagnosis of NASH is only possible with a liver biopsy indicative of steatosis, hepatocyte ballooning, and lobular inflammation.⁷ However, the use of liver biopsy is restricted by cost, patient refusal, and requirement for specialist pathologist interpretation.⁸ Several non-invasive approaches may assist with the diagnosis of NASH or suspected NASH and liver fibrosis, including predictive models, serum biomarkers, and imaging techniques.⁹

The primary goals of NASH management are to reduce progression to cirrhosis or hepatocellular carcinoma and to reduce mortality due to NASH.⁷ Currently, there are no NASH-specific therapies approved by the Food and Drug Administration (FDA) or European Medicines Agency (EMA), although many therapies are currently in development.¹⁰ Any pharmacotherapies used in NASH are off-label and include insulin sensitizers, antioxidants such as vitamin E, and lipid-lowering agents.^{7,11} The most established approach to NASH management currently consists of weight loss through a combination of diet and exercise^{12,13} though weight loss through the use of glucagon-like peptide-1 (GLP-1) receptor agonists has also been found to be effective.¹⁴

A prospective study conducted in Cuba on 293 patients with histologically proven NASH revealed that the greatest rates of reduction in fatty liver disease activity score, fibrosis regression, and NASH resolution occurred in patients with weight loss $\geq 10\%$.¹² However, initiating and sustaining the necessary lifestyle changes may be challenging.¹⁵

Although clinical practice guidelines developed by European liver, diabetes, and obesity associations are available,⁷ awareness and uptake of these guidelines are limited and management approaches differ between regions and are inconsistent.^{16,17} Recently, the Italian liver, diabetes, and obesity societies have developed clinical practice guidelines for the diagnosis and management of NASH in Italy.^{11,18,19} However, the uptake of these guidelines in real-life practice in Italy has not been examined.

Despite the high prevalence of NAFLD in Italy, there is a lack of real-world data on the NASH patient population and care pathway. A better understanding of the clinical and treatment landscape of NASH in Italy may lead to improved identification, diagnosis, management, and ultimately patient outcomes, reducing the societal and healthcare burden of the disease. We aimed to describe the patient characteristics, diagnostic approaches, referral pathways, disease management approaches, compliance, and healthcare resource utilization for patients with NASH or suspected NASH attending secondary care centers in Italy.

Materials and Methods

Data Source

Data were extracted from the 2018 Adelphi Real World NASH Disease Specific Programme (DSP)[™] conducted in Italy from January to March 2018. DSPs are large, multinational surveys conducted in routine clinical practice that describe patient demographics and clinical presentation, disease management, any associated treatments prescribed, and health-care resource utilization. Although not longitudinal in nature, each survey provides independent cross-sectional insights with retrospective data capture, which can be used to explore patterns over time. The DSP methodology has been previously described,^{20,21} validated,²² demonstrated to be representative and consistent over time,²³ and has been employed in a previous publication on NASH.¹⁶

Patients and Physicians

Specialist physicians (endocrinologists, gastroenterologists without a subspecialty in hepatology, and gastroenterologists with a subspecialty in hepatology) were eligible to participate. For this analysis, physicians were categorized as either i) an endocrinologist or ii) a gastroenterologist (gastroenterologists and gastroenterologists with a subspecialty in

hepatology were grouped). Physicians were required to have been personally responsible for the management and treatment decisions of patients with biopsy-proven or suspected NASH and must have consulted ≥ 10 patients per month. Physicians completed questionnaires for their next five consecutively consulting patients with NASH or suspected NASH, with all types of physicians completing the same standardized questionnaire for all patients. The questionnaire collected information on patient demographics, clinical characteristics, disease management, and patient healthcare resource use.

Eligible patients for analysis had biopsy-confirmed NASH or suspected NASH as identified by the physician, were \geq 18 years of age and were not participating in NASH clinical trials at time of data capture.

Fibrosis severity was categorized by the physician as no or non-advanced fibrosis (F0-F2) or advanced fibrosis (F3-F4) according to liver biopsy or non-invasive tests (NITs) depending on the physician's clinical practice, and according to the physician's routine clinical judgement.

Ethics

The DSP is noninterventional and solely employs retrospective data collection. Data were collected in such a way that patients and physicians could not be identified directly; all data were aggregated and de-identified before receipt for research and publication in scientific journals. Data collection was consistent with the European Pharmaceutical Marketing Research Association guidelines,²⁴ and had ethics approval obtained from Freiburg Ethics Commission International (FEKI; Approval No. 017/1931).

Statistical Analysis

Results are presented as descriptive analysis; mean \pm standard deviation (SD) for continuous variables and frequency (%) for categorical variables are shown. Base sizes may change due to incomplete data or targeted analyses on certain subgroups. Base sizes are reported per variable.

Results

Patients and Physicians

A total of 76 specialist physicians (58 gastroenterologists [15 with subspecialty in hepatology] and 18 endocrinologists) provided data on 380 patients (endocrinologists, 90 patients; gastroenterologists, 290 patients) with NASH or suspected NASH. Physician details are reported in <u>Supplementary Table 1</u>, most physicians (87%) were hospital-based (endocrinologists, 61%; gastroenterologists, 95%). Overall, NASH patients represented 3% and 8% of all patients seen by endocrinologists and gastroenterologists, respectively. Mean patient age was 58.5 ± 11.1 years, and most patients were male (63%). Approximately, a third of patients never smoked (37%). Mean body mass index was 31.8 ± 5.5 kg/m². Obesity (81%) and T2DM (62%) were the two most common comorbidities; most patients (n=267/379, 70%) had ≥ 3 comorbidities. Most patients (61%) had no or non-advanced fibrosis (fibrosis stage F0-F2). Fifty-eight gastroenterologist-managed patients had advanced fibrosis (fibrosis stage F3-F4; 20%) compared with 16 (18%) endocrinologist-managed patients; of these patients with advanced fibrosis, n=35/58 (60%) and n=6/16 (38%) had ever had a liver biopsy, respectively (Table 1).

Clinical Characteristics and Testing at NASH Diagnosis

Mean time since diagnosis was 1.1 ± 1.4 years. Most patients (56%) were diagnosed by a gastroenterologist, with a third (33%) of patients having a physician-confirmed NAFLD diagnosis prior to NASH diagnosis. Most patients (n=212/372, 57%) had no or non-advanced fibrosis at diagnosis. Gastroenterologists reported 26% of patients (n=81/315) to have advanced fibrosis at diagnosis. For endocrinologists, this was 19% of patients (n=11/57). Among patients with advanced fibrosis at time of diagnosis, half (n=40/81) of gastroenterologist-diagnosed patients and 18% (n=2/11) of endocrinologist-diagnosed patients had a liver biopsy.

General weakness (n=166/301, 55%), fatigue (n=134/301, 45%), and sleep disturbance (n=128/306, 43%) were the most common signs reported by all diagnosing physicians at diagnosis. Sleep disturbance was reported for 45% (n=115/

Table I Patient Demographics and Clinical Characteristics

	Total (N=380)				Endocrinologis (n=90)	t	Gastroenterologist ^a (n=290)			
	Total (N=380)	No LB (n=204)	LB (n=176)	Total (n=90)	No LB (n=58)	LB (n=32)	Total (n=290)	No LB (n=146)	LB (n=144)	
Patient age, years, mean ± SD	58.5 ± 11.1	57.8 ± 10.9	59.4 ± 11.2	59.8 ± 8.4	58.8 ± 8.5	61.7 ± 7.9	58.1 ± 11.8	57.4 ± 11.8	58.9 ± 11.8	
Sex, n (%) Male	240 (63)	122 (60)	118 (67)	51 (57)	31 (53)	20 (63)	189 (65)	91 (62)	98 (68)	
Employment status, n (%)										
Working full-time/part-time	178 (47)	91 (45)	87 (49)	43 (48)	28 (48)	15 (47)	135 (47)	63 (43)	72 (50)	
Not working ^b	195 (51)	106 (52)	89 (51)	45 (50)	28 (48)	17 (53)	150 (52)	78 (53)	72 (50)	
Do not know	7 (2)	7 (3)	0 (0)	2 (2)	2 (3)	0 (0)	5 (2)	5 (3)	0 (0)	
Smoking status, n (%)										
Current smoker	78 (21)	45 (22)	33 (19)	17 (19)	11 (19)	6 (19)	61 (21)	34 (23)	27 (19)	
Ex-smoker	131 (34)	57 (28)	74 (42)	30 (33)	15 (26)	15 (47)	101 (35)	42 (29)	59 (41)	
Never smoked	142 (37)	84 (41)	58 (33)	38 (42)	27 (47)	(34)	104 (36)	57 (39)	47 (33)	
Do not know	29 (8)	18 (9)	11 (6)	5 (6)	5 (9)	0 (0)	24 (8)	13 (9)	11 (8)	
Alcohol units consumed per week, mean ± SD	(n=84)	(n=44)	(n=40)	(n=13)	(n=8)	(n=5)	(n=71)	(n=36)	(n=35)	
	5.1 ± 8.4	5.5 ± 10.6	4.7 ± 5.1	7.5 ± 13.4	8.5 ±16.9	6.0 ±5.7	4.6 ±7.2	4.8 ± 8.8	4.5 ± 5.0	
Fibrosis stage at survey ^c , n (%)										
No/non-advanced fibrosis	231 (61)	129 (63)	102 (58)	63 (70)	40 (69)	23 (72)	168 (58)	89 (61)	79 (55)	
Advanced fibrosis	74 (19)	33 (16)	41 (23)	16 (18)	10 (17)	6 (19)	58 (20)	23 (16)	35 (24)	
Unknown	75 (20)	42 (21)	33 (19)	(2)	8 (14)	3 (9)	64 (22)	34 (23)	30 (21)	
BMI at survey, kg/m², mean ± SD	31.8 ± 5.5	32.1 ± 6.1	31.4 ± 4.7	33.5 ± 8.1	34.3 ± 8.5	32.0 ± 4.2	31.3 ± 4.3	31.3 ± 3.7	31.3 ± 4.8	
BMI classification at survey, n (%)										
<30 kg/m ²	116 (31)	58 (28)	58 (33)	20 (22)	12 (21)	8 (25)	96 (33)	46 (32)	50 (35)	
30.0–34.9 kg/m ²	201 (53)	115 (56)	86 (49)	50 (56)	33 (57)	17 (53)	151 (52)	82 (56)	69 (48)	
35.0–39.9 kg/m ²	49 (13)	23 (11)	26 (15)	13 (14)	7 (12)	6 (19)	36 (12)	16 (11)	20 (14)	
≥40.0 kg/m ²	14 (4)	8 (4)	6 (3)	7 (8)	6 (10)	I (3)	7 (2)	2 (1)	5 (3)	
Top five comorbidities, n (%)										
Obesity ^d	308 (81)	166 (81)	142 (81)	81 (90)	53 (91)	28 (88)	227 (78)	113 (77)	114 (79)	
Type 2 Diabetes	235 (62)	132 (65)	103 (59)	70 (78)	47 (81)	23 (72)	165 (57)	85 (58)	80 (56)	
Dyslipidemia	206 (54)	124 (61)	82 (47)	46 (51)	35 (60)	11 (34)	160 (55)	89 (61)	71 (49)	
Hypertension	195 (51)	106 (52)	89 (51)	49 (54)	35 (60)	14 (44)	146 (50)	71 (49)	75 (52)	
Metabolic syndrome	132 (35)	78 (38)	54 (31)	32 (36)	26 (45)	6 (19)	100 (34)	52 (36)	48 (33)	

188

Dovepress

Other selected concomitant conditions, n (%) Sleep disorder ASCVD ^e CKD	8 (3) 99 (26) 9 (2)	75 (37) 59 (29) 6 (3)	43 (24) 40 (23) 3 (2)	22 (24) 15 (17) 2 (2)	15 (26) 12 (21) 2 (3)	7 (22) 3 (9) 0 (0)	96 (33) 84 (29) 7 (2)	60 (41) 47 (32) 4 (3)	36 (25) 37 (26) 3 (2)
Number of comorbidities ^f , n (%)									
l	16 (4)	10 (5)	6 (3)	0 (0)	0 (0)	0 (0)	16 (6)	10 (7)	6 (4)
	44 (12)	18 (9)	26 (15)	5 (6)	4 (7)	I (3)	39 (13)	14 (10)	25 (17)
2	52 (14)	25 (12)	27 (15)	20 (22)	10 (17)	10 (31)	32 (11)	15 (10)	17 (12)
≥3	267 (70)	150 (74)	117 (66)	65 (72)	44 (76)	21 (66)	202 (70)	106 (73)	96 (67)

Notes: ^a Gastroenterologist category also includes gastroenterologists or internists with a subspecialty in hepatology. ^b Defined as patients who are homemakers, on long-term sick leave, retired, students, or unemployed. ^c As assessed and reported by physician. F-stage F0-F2 was considered as no/non-advanced fibrosis and F3-F4 as advanced fibrosis. ^d Selected as comorbidity or stated current BMI \geq 30 kg/m^{2. e} Defined as having at least one of myocardial infarction, peripheral vascular disease, or cerebrovascular disease. ^f Data from 379 patients, data not available from one patient treated by a gastroenterologist.

Abbreviations: ASCVD, atherosclerotic cardiovascular disease; BMI, body mass index; CKD, chronic kidney disease; LB, liver biopsy; SD, standard deviation.

189

257) of gastroenterologist-diagnosed patients and for 30% (n=13/44) of endocrinologist-diagnosed patients. The most common symptoms reported by all physicians at diagnosis were swelling in the legs, ankles, and feet (n=115/301, 38%), swelling of the stomach/abdomen (n=108/301, 36%), and aching/discomfort in upper right abdomen (n=107/301, 36%). Aching/discomfort in upper right abdomen was reported for 36% (n=93/257) of gastroenterologist-diagnosed and for 32% (n=14/44) of endocrinologist-diagnosed patients (Table 2).

Assessment tests commonly conducted for all patients with NAFLD or suspected NASH at NASH diagnosis included cholesterol (high-density lipoprotein, low-density lipoprotein, total), HbA1c, serological tests for viral hepatitis, and evaluation of alcohol intake (thresholds <20 g/day for females and <30 g/day for males, Figure 1A). Notably, the assessment of alcohol intake was conducted for only 56% (n=32/57) of patients diagnosed by endocrinologists (Figure 1A). Ultrasonography was the most common diagnostic test performed by the physicians to diagnose NASH or suspected NASH, accounting for 87% (n=322/372) of cases. The survey did not cover the specific ultrasonography modalities used. Additionally, aspartate aminotransferase (AST) and alanine aminotransferase (ALT) were also commonly used (Figure 1B). The Fibrosis 4 (FIB-4) index and serum Enhanced Liver Fibrosis (ELF) tests were performed for 5% (n=19/372) and 4% (n=15/372) of patients, respectively.

VCTE was performed on 62% (n=230/372) of patients. Liver biopsy was performed on 45% (n=169/372) of patients; 47% (n=148/315) of gastroenterologist-diagnosed patients and 37% (n=21/57) of endocrinologist-diagnosed patients had liver biopsy (Figure 1B). The mean number of tests performed to aid diagnosis was 8.1 ± 4.7 ; gastroenterologists performed a mean of 8.2 ± 4.9 tests compared with 7.4 ± 3.6 tests by endocrinologists (Table 2).

Referral and Current Managing Physician

Most patients (87%) were referred from another physician instead of presenting directly to the current managing physician (Figure 2A). For both specialties, patients were commonly referred to their current managing physician by a primary care physician (PCP) (n=145/334, 43%); 58% (n=47/81) of endocrinologist-managed patients and 39% (n=99/253) of gastroenterologist-managed patients were referred by a PCP (Figure 2B). Overall, 85% (n=106/124) of endocrinologist-managed patients were co-managed for NASH with another physician. For gastroenterologist-managed patients, this was 63% (n=152/240). Physicians involved in co-management with an endocrinologist were mostly PCPs (n=58/106, 55%) or other gastroenterologists (n=54/106, 51%, Supplementary Table 2).

Further Testing After NASH or Suspected NASH Diagnosis

The mean number of tests conducted by physicians following the diagnosis of NASH or suspected NASH was 5.8 ± 4.0 . Managing gastroenterologists performed a mean of 6.0 ± 4.2 tests and managing endocrinologists performed a mean of 5.2 ± 3.3 tests (Table 3). VCTE was performed on 54% of all patients (Figure 1C). Liver biopsy was performed on 10% of all patients; 9% (n=27/290) of patients managed by gastroenterologists and 13% (n=12/90) of patients managed by endocrinologists had a liver biopsy. Overall, FIB-4 index and serum ELF test were performed for 2% and 1% of patients, respectively.

Physician Attitudes for Role of Liver Biopsy to Manage NASH

Overall, physicians sent on average 37% of their NASH patients to liver biopsy as part of NASH diagnosis. Gastroenterologists and endocrinologists sent 39% and 29% of their patients to liver biopsy, respectively (Figure 3A). Among physicians managing patients who had a liver biopsy (n=54), 88% (n=37/42) of gastroenterologists and 50% (n=6/12) of endocrinologists considered liver biopsy as superior for diagnosing NASH when compared with all other testing options (Figure 3B). Among physicians managing patients who had not had a liver biopsy (n=71), the most common reason for not requesting liver biopsy was invasiveness and discomfort (n=49/71, 69%). This reason was reported by gastroenterologists in 74% of cases (n=39/53, Figure 3C). Liver biopsy was not discussed with 61 patients; the most common reason for physicians not discussing liver biopsy was absence of a currently approved treatment for NASH (n=32/61, 52%, Figure 3D). Liver biopsy was discussed with, and refused by, 24% of patients overall (Figure 3E). For these patients, the most common reason for refusing liver biopsy was fear of procedure (n=68/93, 73%, Figure 3F).

	Current managing physician									
	Total (N=380)			Endocrinologist (n=90)			Gastroenterologist ^a (n=290)			
	Total (N=380)	No LB (n=204)	LB (n=176)	Total (n=90)	No LB (n=58)	LB (n=32)	Total (n=290)	No LB (n=146)	LB (n=144)	
Years since diagnosis, mean \pm SD	(n=299) I.I ± I.4	(n=149) 1.0 ± 1.2	(n=150) 1.3 ± 1.6	(n=62) I.I ± I.5	(n=37) 1.0 ± 1.2	(n=25) I.3 ± I.8	(n=237) I.I ± I.4	(n=112) 1.0 ± 1.1	(n=125) 1.2 ± 1.6	
Diagnosing physician, n (%)										
Gastroenterologist	213 (56)	98 (48)	115 (65)	13 (14)	9 (16)	4 (13)	200 (69)	89 (61)	(77)	
Gastroenterologist ^a	102 (27)	64 (31)	38 (22)	28 (31)	18 (31)	10 (31)	74 (26)	46 (32)	28 (19)	
Endocrinologist	57 (15)	36 (18)	21 (12)	46 (51)	28 (48)	18 (56)	(4)	8 (5)	3 (2)	
PCP	2 (1)	I (0)	1 (12)	0 (0)	0 (0)	0 (0)	2 (1)	I (I)	I (I)	
Other	2 (1) 3 (1)	2 (1)	I (I)	2 (2)	2 (3)	0 (0)	I (0)	0 (0)		
Other Do not know	3 (I) 3 (I)	2 (1) 3 (1)	0 (0)	2 (2) (1)	2 (3) I (2)	0 (0)	2 (1)	2 (1)	0 (0)	
	J (1)	5 (1)	0 (0)	. (.)	. (=)	• (•)	- (.)	- (.)	0 (0)	
NALFD diagnosis prior to suspected NASH/NASH diagnosis, n (%)	105 (22)	50 (20)	(= (20)			0 (20)				
Yes	125 (33)	58 (28)	67 (38)	34 (38)	25 (43)	9 (28)	91 (31)	33 (22)	58 (40)	
		Diagnosing physician								
	Total (N=372 ^b)			Endocrinologist (n=57)			Gastroenterologist ^a (n=315)			
	Total (N=372)	No LB (n=198)	LB (n=174)	Total (n=57)	No LB (n=36)	LB (n=21)	Total (n=315)	No LB (n=162)	LB (n=153)	
Fibrosis stage at diagnosis ^c , n (%)										
No/non-advanced fibrosis	212 (57)	103 (52)	109 (63)	43 (75)	25 (69)	18 (86)	169 (54)	78 (48)	91 (59)	
	. ,	50 (25)	42 (24)	11 (19)	9 (25)	2 (10)	81 (26)	41 (25)	40 (26)	
Advanced fibroris	92 (25)				7 (23)	2 (10)	01 (20)		22 (14)	
	92 (25) 68 (18)					1 (5)	65 (21)	43 (27)		
Unknown	92 (25) 68 (18)	45 (23)	23 (13)	3 (5)	2 (6)	I (5)	65 (21)	43 (27)	22 (14)	
Unknown	68 (18)	45 (23)	23 (13)	3 (5)	2 (6)					
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n	68 (18) 301 ^d	45 (23)	23 (13)	3 (5)	2 (6)	18	257	130	127	
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n General weakness	68 (18) 301 ^d 166 (55)	45 (23) 156 88 (56)	23 (13) 145 78 (54)	3 (5) 44 27 (61)	2 (6) 26 17 (65)	18 10 (56)	257 139 (54)	30 7 (55)	127 68 (54)	
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n General weakness Fatigue	68 (18) 301 ^d 166 (55) 134 (45)	45 (23) 156 88 (56) 72 (46)	23 (13) 145 78 (54) 62 (43)	3 (5) 44 27 (61) 20 (45)	2 (6) 26 17 (65) 14 (54)	18 10 (56) 6 (33)	257 139 (54) 114 (44)	130 71 (55) 58 (45)	127 68 (54) 56 (44)	
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n General weakness	68 (18) 301 ^d 166 (55)	45 (23) 156 88 (56)	23 (13) 145 78 (54)	3 (5) 44 27 (61)	2 (6) 26 17 (65)	18 10 (56)	257 139 (54)	30 7 (55)		
Fatigue	68 (18) 301 ^d 166 (55) 134 (45)	45 (23) 156 88 (56) 72 (46)	23 (13) 145 78 (54) 62 (43)	3 (5) 44 27 (61) 20 (45)	2 (6) 26 17 (65) 14 (54)	18 10 (56) 6 (33)	257 139 (54) 114 (44)	130 71 (55) 58 (45)	127 68 (54) 56 (44)	
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n General weakness Fatigue Sleep disturbance	68 (18) 301 ^d 166 (55) 134 (45) 128 (43)	45 (23) 156 88 (56) 72 (46) 75 (48)	23 (13) 145 78 (54) 62 (43) 53 (37)	3 (5) 44 27 (61) 20 (45) 13 (30)	2 (6) 26 17 (65) 14 (54) 10 (38)	18 10 (56) 6 (33) 3 (17)	257 139 (54) 114 (44) 115 (45)	130 71 (55) 58 (45) 65 (50)	127 68 (54) 56 (44) 50 (39) 39 (31)	
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n General weakness Fatigue Sleep disturbance Swelling in the legs, ankles, feet	301 ^d 166 (55) 134 (45) 128 (43) 115 (38)	45 (23) 156 88 (56) 72 (46) 75 (48) 73 (47)	23 (13) 145 78 (54) 62 (43) 53 (37) 42 (29)	3 (5) 44 27 (61) 20 (45) 13 (30) 17 (39)	2 (6) 26 17 (65) 14 (54) 10 (38) 14 (54)	18 10 (56) 6 (33) 3 (17) 3 (17)	257 139 (54) 114 (44) 115 (45) 98 (38)	130 71 (55) 58 (45) 65 (50) 59 (45)	127 68 (54) 56 (44) 50 (39)	
Unknown Top six physician-reported signs/symptoms prior to/at diagnosis, n (%) n General weakness Fatigue Sleep disturbance Swelling in the legs, ankles, feet Swelling of the stomach/abdomen	301 ^d 166 (55) 134 (45) 128 (43) 115 (38) 108 (36)	45 (23) 156 88 (56) 72 (46) 75 (48) 73 (47) 60 (38)	23 (13) 145 78 (54) 62 (43) 53 (37) 42 (29) 48 (33)	3 (5) 44 27 (61) 20 (45) 13 (30) 17 (39) 16 (36)	2 (6) 26 17 (65) 14 (54) 10 (38) 14 (54) 14 (54)	18 10 (56) 6 (33) 3 (17) 3 (17) 2 (11)	257 139 (54) 114 (44) 115 (45) 98 (38) 92 (36)	130 71 (55) 58 (45) 65 (50) 59 (45) 46 (35)	127 68 (54) 56 (44) 50 (39) 39 (31) 46 (36)	

Notes: ^a Gastroenterologist category also includes gastroenterologists with a subspecialty in hepatology. ^b All physicians who conducted tests at diagnosis of NASH or suspected NASH. ^c As assessed and reported by physician. F-stage F0-F2 was considered as no/non-advanced fibrosis and F3-F4 as advanced fibrosis. ^d Lower base as only data from patients with current signs or symptoms were captured. ^e Include the following: imaging tests (computed tomography, Magnetic resonance imaging, Proton magnetic resonance spectroscopy and ultrasonography), blood tests (aspartate aminotransferase [AST], alanine aminotransferase [ALT], AST/ALT ratio, alkaline phosphatase, gamma-glutamyltransferase, platelets, international normalized ratio, serum albumin, and total bilirubin) and liver-specific (actiTest, BARD Score, fatty liver index, FIB-4 index, VCTE, fibrotest, liver biopsy, NAFLD activity score, NAFLD fibrosis score).

Abbreviations: LB, liver biopsy; NAFLD, non-alcoholic fatty liver disease; NASH, non-alcoholic steatohepatitis; PCP, primary care physician; SD, standard deviation.

161



Figure I Top ten elimination/assessment tests (\mathbf{A}) and tests for NASH or suspected NASH (\mathbf{B}) conducted to aid diagnosis by diagnosing physician. Top ten tests for NASH or suspected NASH conducted after NASH or suspected NASH diagnosis by managing physician (\mathbf{C}). For all panels, percentage values indicated are for all patients for that particular category. The gastroenterologist category also included gastroenterologists with a subspecialty in hepatology. Tests are categorized by frequency of total percentage, from lowest to highest.

Abbreviations: ALT, alanine aminotransferase; ALP, alkaline phosphatase; AST, aspartate aminotransferase; GGT, γ-glutamyltransferase; HbAIC, glycated hemoglobin; HDL, highdensity lipoprotein; INR, international normalized ratio; LDL, low-density lipoprotein; T2DM, type 2 diabetes mellitus; VCTE, vibration-controlled transient elastography.



Figure 2 (A) Percentage of patients referred either directly to managing physician or referred by another physician. (B) Specialty of physician who referred patient. Other infectious disease specialist, cardiologist, specialist nurse, radiologist, surgeon, neurologist, and not specified. For both panels, current managing physician is indicated in the Y-axis label. The specialty of the referring physician is shown by the different colors as defined below panel B. The gastroenterologist category also included gastroenterologists with a subspecialty in hepatology.

Abbreviations: HCP, healthcare provider; PCP, primary care physician.

Patient Management and Treatment After NASH or Suspected NASH Diagnosis

Nearly two-thirds of all patients (58%) were receiving ≥ 1 treatment to manage NASH (weight-loss treatment, guideline-recommended NASH treatment, or treatment for associated conditions). Fifty-four (14%) patients were receiving or had received weight-loss treatment. For all patients, the most common guideline-recommended NASH treatments were vitamin E (13%) and GLP-1 receptor agonists (13%).^{7,11,25} GLP-1 receptor agonists were prescribed to 28% (n=25/90) of patients managed by endocrinologists and only 9% (n=25/290) of patients managed by gastroenterologists. Overall, 5% of patients were currently receiving pioglitazone to manage their NASH or suspected NASH; this was prescribed to 11% (n=10/90) of patients managed by endocrinologists and 3% (n=10/290) of patients managed by gastroenterologists (Table 3). Overall, 31% of patients were using statins, of whom only 3% were clinically diagnosed with T2DM (Supplementary Table 2). In addition, statins were prescribed for 27% of endocrinologist-managed patients and 32% of gastroenterologist patients.

A total of 62% and 81% of patients with NASH or suspected NASH were also diagnosed with T2DM and obesity, respectively. Among patients with T2DM or obesity (BMI \geq 30 kg/m²), 66% (n=154/235) and 62% (n=192/308, 62%) were receiving \geq 1 treatment to manage NASH, respectively. The corresponding values for those without T2DM or without obesity were 45% (n=65/145) and 38% (n=27/72), respectively. GLP-1 receptor agonists were the most common guideline-recommended NASH treatment prescribed to 20% (n=46/235) of patients with T2DM and 3% (n=4/145) of patients without T2DM. Vitamin E was prescribed to patients with T2DM (n=27/235, 11%) or without T2DM (n=22/145, 15%). Pioglitazone was prescribed to 11% (n=8/70) of endocrinologist-managed patients with T2DM to target NASH or



Figure 3 (A) Physician-estimated proportion of patients sent to liver biopsy. (B) Reasons why physicians sent patient to liver biopsy. Percentage values indicated are for all physicians for that particular category. (C) Top ten reasons why physicians did not send patient to liver biopsy. Percentage values indicated are for all physicians for that particular category. (D) Reasons why physician did not discuss liver biopsy with patient. Percentage values refer to proportions of patients treated by all physicians for that particular category. (E) Discussion of liver biopsy with patient. (F) Reasons why patients refused liver biopsy. Percentage values refer to proportions of patients treated by all physicians for that particular category. Physician survey data shown in panels A–C; data from physician-reported patient record forms shown in panels D–F.

Table 3 Patient Management and Treatment After NASH or Suspected NASH Diagnosis

	Total	Current managing physician			
	(N=380)	Endocrinologist (n=90)	Gastroenterologist ^a (n=290)		
Receiving ≥ 1 treatment to manage NASH at time of survey, n (%) ^a	219 (58)	61 (68)	158 (54)		
Treatments received for NASH at time of survey as recommended by Italian					
guidelines, n (%) ^b					
Any guideline-recommended treatment	106 (28)	40 (44)	66 (23)		
Vitamin E	49 (13)	10 (11)	39 (13)		
GLP-I	50 (13)	25 (28)	25 (9)		
Pioglitazone	20 (5)	10 (11)	10 (3)		
Treatments received for NASH-associated conditions, n (%) ^c	179 (47)	52 (58)	127 (44)		
Any anti-hypertensive treatment ^d	237 (65)	58 (64)	179 (66)		
Any anti-diabetic treatment ^e	148 (39)	42 (47)	106 (37)		
Metformin	141 (37)	40 (44)	101 (35)		
DPP-4	12 (3)	4 (4)	8 (3)		
SGLT-2	12 (3)	4 (4)	8 (3)		
Any weight-loss treatment ^f	54 (14)	20 (22)	34 (12)		
Prescription anti-obesity medication	23 (6)	11 (12)	12 (4)		
Non-prescription weight-loss drug	38 (10)	10 (11)	28 (10)		
Previous weight-loss surgery	4 (1)	2 (2)	2 (1)		
Top five reasons why patient has never received drug treatment for NASH,	n=153	n=29	n=124		
n (%)					
No approved medicines	81 (53)	12 (41)	69 (56)		
Recent diagnosis/currently undergoing evaluation	35 (23)	9 (31)	26 (21)		
Patient refused treatment	24 (16)	3 (10)	21 (17)		
Cost/insurance	17 (11)	2 (7)	15 (12)		
Patient has other health problems	6 (4)	2 (7)	4 (3)		
Top five reasons for reduced drug compliance, n (%) ^g	n= 79	n=31	n=48		
Change of routine	37 (47)	12 (39)	25 (52)		
Does not always carry medication with them	18 (23)	5 (16)	13 (27)		
Does not feel ill, medication unnecessary	16 (20)	8 (26)	8 (17)		
Confused as so many drugs in total	15 (19)	8 (26)	7 (15)		
Lack of family/friend support	12 (15)	I (3)	11 (23)		
Tests used after diagnosis ^h					
Mean ± SD	5.8 ± 4.0	5.2 ± 3.3	6.0 ± 4.2		

Notes: ^a Defined as having at least one of the following: currently on treatment with a guideline-recommended treatment (vitamin E, pioglitazone, or GLP-1), currently on anti-diabetic treatment (metformin, DPP-4, or SGLT-2), or currently on weight-loss treatment (had weight-loss surgery, currently on prescription weight-loss drug, or currently on non-prescription (over the counter) weight-loss drug). ^b Defined as currently on treatment with a guideline-recommended treatment (vitamin E, pioglitazone, or GLP-1). Use of GLP-1 is recommended in the context of T2DM. Refer to: Eat Weight Disord 2022;27:1603–19. ^c Defined as having at least one of the following: currently on anti-diabetic drug (metformin, DPP-4, or SGLT-2), or currently on weight-loss treatment (had weight-loss surgery, currently on a prescription weight-loss drug, or currently on non-prescription (over the counter) weight-loss drug). ^d Defined as currently on anti-diabetic treatment (and guideline-recommended treatment (vitamin E, pioglitazone, or GLP-1), inhibitors, angiotensin-converting-enzyme inhibitors, alpha blockers, beta blockers, calcium channel blockers, diuretics, renin inhibitors, assolilators or other anti-hypertensive medication). ^e Defined as currently on anti-diabetic treatment (metformin, DPP-4, or SGLT-2). Note: excludes GLP-1 as included as guideline treatment for NASH. ^f Defined as having at least one of the following: had weight-loss surgery, currently on prescription weight-loss drug, or currently on non-prescription (over the counter) weight-loss drug. ^b Includes the following: imaging tests (computed tomography, magnetic resonance imaging, proton magnetic resonance spectroscopy and ultrasonography), blood tests (aspartate aminotransferase (AST), alanine aminotransferase (ALT), AST/ALT ratio, alkaline phosphatase, gamma-glutamyltransferase, platelets, international normalised ratio, serum albumin and total bilirubin) and liver specific (actiTest, BARD Score, fatty liver index, FIB4 index, VCTE, FibroTest, liver

suspected NASH compared with 4% (n=7/165) of patients managed by gastroenterologists (<u>Supplementary Table 3</u>). The most common reason for patients not receiving drug treatment for NASH was the absence of approved treatments (n=81/153, 53%). The most common reason for lack of patient compliance with drug treatment was patient change of daily routine (n=37/79, 47%) (Table 3).

Healthcare Resource Use

Among all patients with available hospitalization data (n=231), 15% had been hospitalized in the past 12 months for any reason. Of these, 51% were hospitalized for a NASH-associated condition and 37% for NASH. A total of 16% (n=30/ 184) gastroenterologist-managed patients and 11% (n=5/47) of endocrinologist-managed patients were hospitalized for any reason in the last 12 months (Supplementary Table 4).

A total of 28% (n=8/29) of patients with advanced fibrosis were hospitalized in the last 12 months; 63% (n=5/8) of patients with advanced fibrosis were hospitalized in the last 12 months for a NASH-associated condition. The corresponding values for those with no or non-advanced fibrosis were 18% (n=25/142) and 48% (n=12/25), respectively (Supplementary Table 4).

Discussion

This real-world survey of patients with NASH or suspected NASH in Italy revealed that these patients are often obese and have a high burden of comorbidities. Ultrasonography was performed on most patients for diagnosis, with liver biopsy performed on slightly less than half of patients. VCTE was a NIT commonly performed by diagnosing and managing physicians. Common clinical investigations or biochemical tests performed included cholesterol, HbA1c, serological tests for viral hepatitis, and assessment of alcohol intake.

Most patients were referred by another physician instead of presenting directly to the current managing physician for NASH management. Although nearly two-thirds of all patients were receiving at least one treatment to manage NASH, treatment according to current guidelines appears suboptimal. Pathologic diagnosis remains a key standard for establishing diagnosis of NAFLD, with radiological modalities also being widely used. However, radiological modalities such as ultrasonography, computerized tomography and magnetic resonance imaging lack the ability to detect or characterize NASH from steatosis alone, or are able to detect the different forms of NAFLD.^{26,27}

These modalities can also produce nonspecific findings, which may lead to a significant variability in interpretation among radiologists. It is therefore important to be able to distinguish a clear prognosis of patients with NAFLD, with better clinical, radiological and serological markers of fibrosis, which highly supports the role of liver biopsy in staging liver disease in NAFLD.²⁸ In this study we do not specify which radiology modality is used by our physicians. In addition, this study contains descriptive analysis, and as such we did not compute any data to identify any independent predictor of the decision of performing liver biopsy.

Gastroenterologists tended to evaluate alcohol intake more frequently than endocrinologists. To the best of our knowledge, data on the attitudes towards alcohol assessment among gastroenterologists and endocrinologists in the context of NASH management is limited. It may be possible that liver health and associated assessments are more likely to be standard of care among gastroenterologists. The emphasis on liver health and alcohol assessment may also be reflected, in the proportion (approximately a quarter) of physicians with a subspecialty in hepatology, among the gastroenterologists recruited in this study. As >60% of the patients in this study had T2DM, it is possible that the managing endocrinologists were more focused on addressing T2DM and were less aware of or focused on liver health.

Patients currently managed by endocrinologists were frequently referred by a PCP, suggesting that there is already an effective referral pathway in Italy from PCPs to endocrinologists, but not to gastroenterologists. Endocrinologists may be more likely to reach the general population through their connection with PCPs. Another possibility may be that PCPs are referring to endocrinologists due to the high prevalence of T2DM in this population, as there may be a lack of liver indicators due to the asymptomatic nature of NASH. Hence, through routine T2DM assessment, endocrinologists may be subsequently identifying suspected NASH. These results indicate that a similar pathway between gastroenterologists and PCPs in Italy has not been established and development of such a pathway is warranted.

The FIB-4 for liver fibrosis is recommended as a first-line non-invasive assessment of liver fibrosis for PCPs and endocrinologists due to its simplicity, high negative predictive value, and minimal cost.^{25,29–31} Despite this recommendation, FIB-4 was used for <10% of all patients by both diagnosing and managing physicians in this analysis. The minimal use of FIB-4 may be due to requirement for manual calculation or lack of awareness of the assessment or relevant cutoffs for NASH. Use of FIB-4 may be increased by automating and reimbursing this assessment (FIB-4 is currently only reimbursed in Italy in the context of T2DM) and providing cut-off values on every lab report. Reimbursement and provision of cut-off values should be considered for the serum ELF test also, which was used for <10% of patients in this analysis. A study on FIB-4 in the community setting suggested that FIB-4 can facilitate early identification of patients at high risk for advanced liver fibrosis.³² A systematic review and meta-analysis of ELF in patients with NAFLD revealed a high negative predictive value of ELF, suggesting that ELF can exclude advanced fibrosis in this population.³³ Accordingly, increased use of FIB-4 and ELF may assist in the risk stratification of patients in a primary care setting and expedite referral to specialist management if appropriate.

Current Italian guidelines state that NITs do not have acceptable accuracy for NASH diagnosis and that liver biopsy is the reference standard.¹¹ The proportion of patients (nearly 50%) who received a liver biopsy to aid diagnosis was perceived to be high for real-world clinical practice, with the possibility that this procedure could have been performed as part of a clinical trial and not in real-world clinical practice, although this could not be verified. It was also not possible to determine if liver biopsy was performed before or after NITs. Broad use of liver biopsy in real-world settings for NASH diagnosis may be restricted by cost, patient refusal, and requirement for specialist interpretation.⁸ The current Italian guidelines also state that FIB-4 and VCTE can be used instead of liver biopsy to exclude patients at high risk of advanced fibrosis.¹¹ Half of diagnosing and managing physicians used VCTE in this analysis. VCTE was performed on over two-thirds of patients diagnosed by endocrinologists, although the rationale for VCTE and whether it was already performed or if it was requested by the diagnosing endocrinologist are unknown. An increased awareness of, access to, and reimbursement for VCTE (along with FIB-4 and ELF, discussed above) may improve risk stratification and referral.

Approximately a third of patients in our cohort received statins. Patients with NAFLD are frequently characterized by the presence of cardiometabolic diseases including type II diabetes, obesity and/or dyslipidemia (collectively known as metabolic syndrome), which may both hasten the progression to more severe forms of NAFLD and increase cardiovascular risk in NAFLD patients.^{34,35} For this reason, NAFLD patients should be prescribed statins to reduce cardiovascular risk. However, due to concerns over statin treatment safety in NAFLD patients,^{36–38} many patients are not given statins, which leads to increased cardiovascular risks.³⁹ However, evidence supports that statins not only reduce cardiovascular risk but can display beneficial effects on the liver.^{40–43}

There is currently no available FDA- or EMA-approved therapy specific for NASH. Although specific treatment guidelines (including Italy-specific guidelines)^{7,8,11,25} are available for NASH, the results from this analysis suggest that uptake of these guidelines in current practice remains suboptimal in Italy. Vitamin E and pioglitazone were used by <15% and <10% of patients, respectively. Despite a recommendation against its use in the context of T2DM, vitamin E was used by 11% of patients with T2DM.⁸ GLP-1 therapies were used by 20% of patients with T2DM. While the challenges associated with implementation of clinical practice guidelines into real-world practice have been acknowledged,⁴⁴ increased use of current guideline-recommended treatments for NASH may improve outcomes in this population until approved NASH treatments are available.

This analysis has some limitations. Sex (biological structure) and gender (social constructs) differences may be key modifiers of health, disease and medicine.⁴⁵ More specifically, it has been recognized that sex differences do exist for NAFLD,⁴⁶ including sex hormones/menopausal status, age and other reproductive factors. However, the current cohort is not of sufficient size to allow sub-group analysis with sufficient statistical power. Future work to address these factors may help in providing better targeted treatment to diverse patient groups.

Participating physicians were more likely to have an interest in NASH management and as such are currently identifying patients with NASH or suspected NASH. Accordingly, their patients may receive different treatment or management options than those managed by physicians with less of a specialist interest. This may be reflected in the preponderance of hospital-based physicians in this analysis. When compared with gastroenterologists, VCTE was commonly performed for patients diagnosed or managed by endocrinologists, possibly suggesting that the

endocrinologists included in this analysis had an interest in NASH. Despite this finding, there may still be limited awareness of NAFLD among endocrinologists, given the recent recommendation that endocrinologists and associated societies should develop strategies to address NAFLD.⁴⁷ NASH patients were identified based on the judgement of the respondent physician and not on a formalized diagnostic checklist. However, this is representative of physician's real-world classification of patients. Recall bias, a common limitation of surveys, may have affected physician responses to the questionnaires. However, physicians had access to patient medical records or charts; recall bias is unlikely a problem. Some data, such as fibrosis stage, may be unknown due to testing being done by a previous treating physician, prior to the start of patient management by the physician reporting on the patients.

Conclusions

In conclusion, this real-world study provides valuable insights into the patient and clinical characteristics and management approaches of patients with suspected NASH or established NASH in Italy. Increased awareness and uptake of currently available guidelines may optimize resource use and improve outcomes in this highly comorbid population.

Abbreviations

ALT, alanine aminotransferase; AST, aspartate aminotransferase; BMI, body mass index; DSP, Disease Specific Programme; EMA, European Medicines Agency; ELF, Enhanced liver fibrosis; FEKI, Freiburg Ethics Commission International; FIB-4, Fibrosis 4; FDA, Food and Drug Administration; GLP-1, Glucagon-like peptide; HDL, High density lipoprotein; LDL, low density lipoprotein; NASH, non-alcoholic steatohepatitis; NAFLD, Non-alcoholic fatty liver disease; NITs, Non-invasive tests; PCP, Primary care physician; SD, Standard deviation; T2DM, Type 2 diabetes mellitus; VCTE, vibration-controlled transient elastography.

Data Sharing Statement

All data, ie, methodology, materials, data and data analysis, that support the findings of this survey are the intellectual property of Adelphi Real World. All requests for access should be addressed directly to Andrea Leith at Andrea. Leith@adelphigroup.com. Andrea Leith is an employee of Adelphi Real World.

Ethics Approval and Informed Consent

The DSP is noninterventional and solely employs retrospective data collection. Data were collected in such a way that patients and physicians could not be identified directly; all data were aggregated and de-identified before receipt for research and publication in scientific journals. Data collection was consistent with the European Pharmaceutical Marketing Research Association guidelines and had ethics approval obtained from Freiburg Ethics Commission International (FEKI; Approval No. 017/1931). Patients provided informed consent to take part in the survey using a checkbox.

Acknowledgments

Medical writing support under the guidance of the authors was provided by Derek Ho, PhD (ScriboMedica Ltd) on behalf of Adelphi Real World in accordance with Good Publication Practice (GPP3) guidelines.⁴⁸

Author Contributions

Elisabetta Bugianesi was responsible for clinical oversight and guidance as lead author.

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding

The analysis described here used data from the Adelphi Real World NASH Disease Specific Programme[™] (DSP). Data collection for the NASH DSP was undertaken by Adelphi Real World as part of an independent survey. The DSP is a wholly owned Adelphi Real World product. Novo Nordisk is one of multiple subscribers to this DSP. Novo Nordisk did not influence the original survey through either contribution to the design of questionnaires or data collection, and publication was not contingent on the subscriber's approval or censorship of the manuscript.

Disclosure

EB is on advisory board of Boehringer Ingelheim, Gilead, Inventiva, MSD and Novo Nordisk and received grant from Gilead Sciences. LM provides advisor/consultancy services to Alfa-Sigma, Boehringer Ingelheim, BMS, Echosens, Galmed, Gilead Sciences, IBSA, Intercept, MEDA, MyGenomics, MSD-Merck Sharp & Dohme, Novartis, Novo Nordisk, Pfizer, Promethera, Resalis Pharma, Rottapharm-Madaus, Siemens Healthineers and Synageva; and provides support for research for Gilead, Intercept and Siemens Healthineers. GD, KG, MM and PP are employees of Novo Nordisk and own stocks in the company. AL and VH are employees of Adelphi Real World. The authors report no other conflicts of interest in this work.

References

- 1. Younossi ZM, Golabi P, Paik JM, Henry A, Van Dongen C, Henry L. The global epidemiology of nonalcoholic fatty liver disease (NAFLD) and nonalcoholic steatohepatitis (NASH): a systematic review. *Hepatology*. 2023;77(4):1335–1347. doi:10.1097/HEP.00000000000004
- 2. Petta S, Di Marco V, Pipitone RM, et al. Prevalence and severity of nonalcoholic fatty liver disease by transient elastography: genetic and metabolic risk factors in a general population. *Liver Int.* 2018;38(11):2060–2068. doi:10.1111/liv.13743
- 3. Miele L, Grattagliano I, Lapi F, et al. Non-alcoholic fatty liver disease and the risk of fibrosis in Italian primary care services: GPS-NAFLD study: GPS-NAFLD Study. *Liver Int.* 2022;42(12):2632–2645. doi:10.1111/liv.15443
- Estes C, Anstee QM, Arias-Loste MT, et al. Modeling NAFLD disease burden in China, France, Germany, Italy, Japan, Spain, United Kingdom, and United States for the period 2016-2030. J Hepatol. 2018;69(4):896–904. doi:10.1016/j.jhep.2018.05.036
- 5. Nd AM. Non-alcoholic fatty liver disease, an overview. Integr Med. 2019;18(2):42-49.
- 6. Cook NS, Nagar SH, Jain A, et al. Understanding patient preferences and unmet needs in non-alcoholic steatohepatitis (NASH): insights from a qualitative online bulletin board study. *Adv Ther.* 2019;36(2):478–491. doi:10.1007/s12325-018-0856-0
- 7. European Association for the Study of the L, European Association for the Study of D, European Association for the Study of O. EASL-EASD-EASO Clinical Practice Guidelines for the management of non-alcoholic fatty liver disease. *J Hepatol*. 2016;64(6):1388–1402. doi:10.1016/j.jhep.2015.11.004
- Chalasani N, Younossi Z, Lavine JE, et al. The diagnosis and management of nonalcoholic fatty liver disease: practice guidance from the American association for the study of liver diseases. *Hepatology*. 2018;67(1):328–357. doi:10.1002/hep.29367
- 9. Younossi ZM, Loomba R, Anstee QM, et al. Diagnostic modalities for nonalcoholic fatty liver disease, nonalcoholic steatohepatitis, and associated fibrosis. *Hepatology*. 2018;68(1):349–360. doi:10.1002/hep.29721
- 10. Fraile JM, Palliyil S, Barelle C, Porter AJ, Kovaleva M. Non-alcoholic steatohepatitis (NASH) a review of a crowded clinical landscape, driven by a complex disease. *Drug Des Devel Ther.* 2021;15:3997–4009. doi:10.2147/DDDT.S315724
- 11. Associazione Italiana per lo Studio del Fegato SIdD, Societa Italiana dO. Non-alcoholic fatty liver disease in adults 2021: a clinical practice guideline of the Italian association for the study of the liver (AISF), the Italian society of diabetology (SID) and the Italian society of obesity (SIO). *Eat Weight Disord*. 2022;27(5):1603–1619. doi:10.1007/s40519-021-01287-1
- Vilar-Gomez E, Martinez-Perez Y, Calzadilla-Bertot L, et al. Weight loss through lifestyle modification significantly reduces features of nonalcoholic steatohepatitis. *Gastroenterology*. 2015;149(2):367–378. doi:10.1053/j.gastro.2015.04.005
- 13. Romero-Gomez M, Zelber-Sagi S, Trenell M. Treatment of NAFLD with diet, physical activity and exercise. J Hepatol. 2017;67(4):829-846. doi:10.1016/j.jhep.2017.05.016
- 14. Nevola R, Epifani R, Imbriani S, et al. GLP-1 receptor agonists in non-alcoholic fatty liver disease: current evidence and future perspectives. Int J Mol Sci. 2023;24(2):1703. doi:10.3390/ijms24021703
- 15. Corey KE, Rinella ME. Medical and surgical treatment options for nonalcoholic steatohepatitis. *Dig Dis Sci.* 2016;61(5):1387–1397. doi:10.1007/s10620-016-4083-8
- 16. Anstee QM, Hallsworth K, Lynch N, et al. Real-world management of non-alcoholic steatohepatitis differs from clinical practice guideline recommendations and across regions. JHEP Rep. 2022;4(1):100411. doi:10.1016/j.jhepr.2021.100411
- 17. Schattenberg JM, Anstee QM, Caussy C, Bugianesi E, Popovic B. Differences between current clinical guidelines for screening, diagnosis and management of nonalcoholic fatty liver disease and real-world practice: a targeted literature review. *Expert Rev Gastroenterol Hepatol.* 2021;15 (11):1253–1266. doi:10.1080/17474124.2021.1974295
- 18. Italian Association for the Study of the L. AISF position paper on nonalcoholic fatty liver disease (NAFLD): updates and future directions. *Dig Liver Dis*. 2017;49(5):471–483. doi:10.1016/j.dld.2017.01.147
- 19. Loria P, Adinolfi LE, Bellentani S, et al. Practice guidelines for the diagnosis and management of nonalcoholic fatty liver disease. A decalogue from the Italian Association for the study of the liver (AISF) expert committee. *Dig Liver Dis.* 2010;42(4):272–282. doi:10.1016/j.dld.2010.01.021
- Anderson P, Benford M, Harris N, Karavali M, Piercy J. Real-world physician and patient behaviour across countries: disease-specific programmes

 a means to understand. *Curr Med Res Opin*. 2008;24(11):3063–3072. doi:10.1185/03007990802457040
- 21. Anderson P, Higgins V, Courcy J, et al. Real-world evidence generation from patients, their caregivers and physicians supporting clinical, regulatory and guideline decisions: an update on disease specific programmes. *Curr Med Res Opin*. 2023;39(12):1707–1715. doi:10.1080/03007995.2023.2279679

- 22. Babineaux SM, Curtis B, Holbrook T, Milligan G, Piercy J. Evidence for validity of a national physician and patient-reported, cross-sectional survey in China and UK: the disease specific programme. *BMJ Open*. 2016;6(8):e010352. doi:10.1136/bmjopen-2015-010352
- 23. Higgins V, Piercy J, Roughley A, et al. Trends in medication use in patients with type 2 diabetes mellitus: a long-term view of real-world treatment between 2000 and 2015. *Diabetes Metab Syndr Obes*. 2016;9:371–380. doi:10.2147/dmso.s120101
- 24. ephmra.org. Code of conduct and adverse event reporting guidelines. [Cited June 28, 2023] 2022.
- 25. Rinella ME, Neuschwander-Tetri BA, Siddiqui MS, et al. AASLD practice guidance on the clinical assessment and management of nonalcoholic fatty liver disease. *Hepatology*. 2023;77(5):1797–1835. doi:10.1097/HEP.0000000000323
- 26. Celle G, Savarino V, Picciotto A, Magnolia MR, Scalabrini P, Dodero M. Is hepatic ultrasonography a valid alternative tool to liver biopsy? Report on 507 cases studied with both techniques. *Dig Dis Sci.* 1988;33(4):467–471. doi:10.1007/BF01536033
- 27. Siegelman ES, Rosen MA. Imaging of hepatic steatosis. Semin Liver Dis. 2001;21(1):71-80. doi:10.1055/s-2001-12930
- 28. Saadeh S, Cammell G, Carey WD, Younossi Z, Barnes D, Easley K. The role of liver biopsy in chronic hepatitis C. *Hepatology*. 2001;33 (1):196–200. doi:10.1053/jhep.2001.20534
- 29. Loomba R, Jain A, Diehl AM, et al. Validation of serum test for advanced liver fibrosis in patients with nonalcoholic steatohepatitis. *Clin Gastroenterol Hepatol.* 2019;17(9):1867–1876. doi:10.1016/j.cgh.2018.11.004
- 30. Cui J, Ang B, Haufe W, et al. Comparative diagnostic accuracy of magnetic resonance elastography vs. eight clinical prediction rules for non-invasive diagnosis of advanced fibrosis in biopsy-proven non-alcoholic fatty liver disease: a prospective study. *Aliment Pharmacol Ther*. 2015;41(12):1271–1280. doi:10.1111/apt.13196
- Younossi ZM, Corey KE, Alkhouri N, et al. Clinical assessment for high-risk patients with non-alcoholic fatty liver disease in primary care and diabetology practices. *Aliment Pharmacol Ther*. 2020;52(3):513–526. doi:10.1111/apt.15830
- 32. Blanco-Grau A, Gabriel-Medina P, Rodriguez-Algarra F, et al. Assessing liver fibrosis using the FIB4 index in the community setting. *Diagnostics*. 2021;11(12):2236. doi:10.3390/diagnostics11122236
- 33. Vali Y, Lee J, Boursier J, et al. Enhanced liver fibrosis test for the non-invasive diagnosis of fibrosis in patients with NAFLD: a systematic review and meta-analysis. J Hepatol. 2020;73(2):252–262. doi:10.1016/j.jhep.2020.03.036
- 34. Karajamaki AJ, Bloigu R, Kauma H, et al. Non-alcoholic fatty liver disease with and without metabolic syndrome: different long-term outcomes. *Metabolism*. 2017;66:55–63. doi:10.1016/j.metabol.2016.06.009
- 35. Stahl EP, Dhindsa DS, Lee SK, Sandesara PB, Chalasani NP, Sperling LS. Nonalcoholic fatty liver disease and the heart: JACC state-of-the-art review. J Am Coll Cardiol. 2019;73(8):948–963. doi:10.1016/j.jacc.2018.11.050
- 36. Adhyaru BB, Jacobson TA. Safety and efficacy of statin therapy. Nat Rev Cardiol. 2018;15(12):757-769. doi:10.1038/s41569-018-0098-5
- 37. Bhardwaj SS, Chalasani N. Lipid-lowering agents that cause drug-induced hepatotoxicity. Clin Liver Dis. 2007;11(3):597-613. doi:10.1016/j. cld.2007.06.010
- 38. Simic I, Reiner Z. Adverse effects of statins myths and reality. Curr Pharm Des. 2015;21(9):1220–1226. doi:10.2174/1381612820666141013134447
- 39. Marrs JC, Kostoff MD. Discontinuation of statins: what are the risks? Curr Atheroscler Rep. 2016;18(7):41. doi:10.1007/s11883-016-0596-z
- 40. Haffner SM. Statin therapy for the treatment of diabetic dyslipidemia. Diabetes Metab Res Rev. 2003;19(4):280-287. doi:10.1002/dmrr.393
- 41. Nascimbeni F, Pellegrini E, Lugari S, et al. Statins and nonalcoholic fatty liver disease in the era of precision medicine: more friends than foes. *Atherosclerosis*. 2019;284:66–74. doi:10.1016/j.atherosclerosis.2019.02.028
- 42. Eslami L, Merat S, Malekzadeh R, Nasseri-Moghaddam S, Aramin H. Statins for non-alcoholic fatty liver disease and non-alcoholic steatohepatitis. *Cochrane Database Syst Rev.* 2013;12:CD008623. doi:10.1002/14651858.CD008623.pub2
- 43. Pastori D, Pani A, Di Rocco A, et al. Statin liver safety in non-alcoholic fatty liver disease: a systematic review and metanalysis. Br J Clin Pharmacol. 2022;88(2):441-451. doi:10.1111/bcp.14943
- 44. Beauchemin M, Cohn E, Shelton RC. Implementation of clinical practice guidelines in the health care setting: a concept analysis. ANS Adv Nurs Sci. 2019;42(4):307–324. doi:10.1097/ANS.0000000000263
- 45. Mauvais-Jarvis F, Bairey Merz N, Barnes PJ, et al. Sex and gender: modifiers of health, disease, and medicine. *Lancet*. 2020;396(10250):565–582. doi:10.1016/S0140-6736(20)31561-0
- 46. Lonardo A, Nascimbeni F, Ballestri S, et al. Sex differences in nonalcoholic fatty liver disease: state of the art and identification of research gaps. *Hepatology*. 2019;70(4):1457–1469. doi:10.1002/hep.30626
- 47. Byrne CD, Targher G. How should endocrinologists diagnose and treat non-alcoholic fatty liver disease? *Lancet Diabetes Endocrinol.* 2022;10 (7):478–480. doi:10.1016/S2213-8587(22)00167-X
- DeTora LM, Toroser D, Sykes A, et al. Good publication practice (GPP) guidelines for company-sponsored biomedical research: 2022 update. Ann Intern Med. 2022;175(9):1298–1304. doi:10.7326/M22-1460

Pragmatic and Observational Research



Publish your work in this journal

Pragmatic and Observational Research is an international, peer-reviewed, open access journal that publishes data from studies designed to reflect more closely medical interventions in real-world clinical practice compared with classical randomized controlled trials (RCTs). The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/ testimonials.php to read real quotes from published authors.

Submit your manuscript here: http://www.dovepress.com/pragmatic-and-observational-research-journal